Author's response to reviews

Title: Rate of complications in scoliosis surgery - a systematic review of the PubMed literature

Authors:

Hans-Rudolf Weiss (hr.weiss@asklepios.com)
Deborah Goodall (Deborahgoodall@nhs.net)

Version: 5 Date: 31 July 2008

Author's response to reviews:

Major Changes:
References - We are unable to access the links at references 136 and 137 can you amend this please: Sorry, I cannot amend because, as said before, the homepage is offline due to server problems so please advise!!!!!!!!!!!!!!!!!!!!!!!

Minor Changes:
Please change the title "Literature" to "References". - Done and also in the text Fig. is changed to figures!
What happens, when a link changes to be inactive after publication??????
Looking forward to an answer this time.
H-R Weiss

Please change all "fig(s)" to "figure(s)" to ensure consistency.